2E2 Tutorial Sheet 3 First Term^1

24 October 2003 1. (2) Using the Laplace transform solve the differential equation f^{′′} + f^′ − 6f = e^{−3t} (1) with boundary conditions f(0) = f^′(0) = 0.

2. (3) Using the Laplace transform solve the differential equation f^{′′} + 6f^′ + 13f = 0 (2) with boundary conditions f(0) = 0 and f^′(0) = 1.

3. (3) Using the Laplace transform solve the differential equation f^{′′} + 6f^′ + 13f = e^t (3) with boundary conditions f(0) = 0 and f^′(0) = 0.

1Conor Houghton, houghton@maths.tcd.ie, see also http://www.maths.tcd.ie/~houghton/2E2.html